Author's response to reviews

Title: International benchmarking of specialty hospitals. A series of case studies on comprehensive cancer centres.

Authors:

Wineke AM Van Lent (w.v.lent@nki.nl)
Relinde D De Beer (rd.d.beer@minvws.nl)
Wim H Van Harten (w.v.harten@nki.nl)

Version: 7 Date: 19 August 2010

Author's response to reviews:

Dear editors,

Thank you for accepting our paper. Herewith we resubmit our paper. We checked the typography of the text and made the changes as requested by the editor:

1. Tables: Please remove bullet points/graphics. All bullet points have been removed
2. Authors’ contributions: Please use initials instead of full name. Changed as requested.
3. Tables- please remove the visible vertical lines from your tables. The layout of the tables has been changed as requested. In the print version the lines have been removed although you can still see them in the online version.
4. Typography: checked as suggested. Minor changes have been made.

Hopefully all the required changes have been made.

On behalf of the authors,

Sincerely,

Wineke van Lent